Author's response to reviews

Title: Effect of troglitazone on tumor growth and pulmonary metastasis development of the mouse osteosarcoma cell line LM8

Authors:

Junichi Aizawa (aizawa@m.ehime-u.ac.jp)
Kenshi Sakayama (kenshi@m.ehime-u.ac.jp)
Setsuya Kamei (setsuya1125@yahoo.ac.jp)
Teruki Kidani (teruteru@m.ehime-u.ac.jp)
Haruyasu Yamamoto (yamaharu@m.ehime-u.ac.jp)
Yoshiaki Norimatsu (ynorimatsu@epu.ac.jp)
Hiroshi Masuno (hmasuno@epu.ac.jp)

Version: 3 Date: 15 February 2010

Author's response to reviews:

Dear Sir/Madam,

We have reduced the volume (228 words) of text presented in the legend of Figure 2. We have changed “compared with the values for untreated cultures/the normal group/the control group” to “compared with the untreated cultures/the normal group/the control group” in legends to Figures 2-5. In addition, we have modified slightly Figure 2. We have replaced "Figures" to "Figure Legends".

Thank you for giving us the opportunity to modify our manuscript.

Best wishes,

Hiroshi Masuno, Ph.D.
Department of Medical Technology,
Faculty of Health Sciences,
Ehime Prefectural University of Health Sciences,
Takooda, Tobe-cho, Iyo-gun, Ehime 791-2101, Japan
Tel. No. +81-89-958-2111
FAX No. +81-89-958-2177
e-mail: hmasuno@epu.ac.jp